Author's response to reviews

Title: Destined to die in hospital? Systematic review and meta-analysis of place of death in haematological malignancy.

Authors:

Debra A Howell (debra.howell@egu.york.ac.uk)
Eve Roman (eve.roman@egu.york.ac.uk)
Helen Cox (helen.cox@egu.york.ac.uk)
Alexandra G Smith (alex.smith@egu.york.ac.uk)
Russell Patmore (russell.patmore@hey.nhs.uk)
Anne C Garry (anne.c.garry@york.nhs.uk)
Martin R Howard (martin.r.howard@york.nhs.uk)

Version: 2  Date: 31 March 2010

Author's response to reviews: see over